Author’s response to reviews

Title: A Combination Intervention Addressing Sexual Risk-Taking Behaviors Among Vulnerable Women in Uganda: Study Protocol for a Cluster Randomized Clinical Trial

Authors:

Fred Ssewamala (fms1@wustl.edu)
OZGE SENSOY BAHAR (ozge.sensoybahar@wustl.edu)
Yesim Tozan (tozan@nyu.edu)
Proscovia Nabunya (nabunyap@wustl.edu)
Larissa Jennings Mayo-Wilson (ljennin6@jhu.edu)
Joshua Kiyingi (joshkiyingi@gmail.com)
Joseph Kagaayi (jkagayi@rhsp.org)
Scarlett Bellamy (bellamys@drexel.edu)
Mary McKay (mary.mckay@wustl.edu)
Susan Witte (ssw12@columbia.edu)

Version: 2 Date: 30 Jul 2019

Author’s response to reviews:

We have added Figure 4 in the text and made sure all figures and the table were referenced in the manuscript.